Author's response to reviews

Title: Seroepidemiology of Toxoplasma gondii infection in pregnant women in a public hospital in northern Mexico

Authors:

Cosme Alvarado-Esquivel (alvaradocosme@yahoo.com)
Antonio Sifuentes-Alvarez (sifual55@hotmail.com)
Sergio Narro-Duarte (alvaradocosme@yahoo.com)
Sergio Estrada-Martinez (sem@mexico.com)
Juan Humberto Diaz-Garcia (tepehuan@dgo.magared.net.mx)
Oliver Liesenfeld (oliver.liesenfeld@charite.de)
Sergio Arturo Martinez-Garcia (sergiomicro@att.net.mx)
Arturo Canales-Molina (alvaradocosme@yahoo.com)

Version: Date: 11 July 2006

Author's response to reviews:


Ref: MS 1985399275102379. "Seroepidemiology of Toxoplasma gondii infection in pregnant women in a public hospital in northern Mexico"

Dear Editor,

Please find a revised manuscript attached. Tables were included at the end of the manuscript. No acknowledgements or funding need mentioning in the manuscript.

Very truly yours,

Dr. Cosme Alvarado Esquivel.
Facultad de Medicina
Avenida Universidad y Fanny Anitua
34000 Durango, Dgo. Mexico.
Tel.: 0052 618 8 128009
Fax: 0052 618 8 130527
Email: alvaradocosme@yahoo.com